Author's response to reviews

Title: Mutation screening of the RNF8, UBC13 and MMS2 genes in Northern Finnish breast cancer families

Authors:
Mikko Vuorela (mikko.vuorela@oulu.fi)
Katri Pylkäs (katri.pylkas@oulu.fi)
Robert Winqvist (robert.winqvist@oulu.fi)

Version: 4 Date: 20 July 2011

Author's response to reviews: see over
Dear Editor,

Thank you for the proofs of our manuscript “Mutation screening of the RNF8, UBC13 and MMS2 genes in Northern Finnish breast cancer families” by Vuorela et al.

As specified below, all requested and other necessary changes have now been done.

Formatting changes to be made:

--------------------------------------

Major revisions (we require the author to make these changes)

Abstract: This should be identical in the manuscript and the submission system. Currently the conclusion is different.

- DONE

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting: Please remove all highlighting from the manuscript.

- DONE

We have also checked the manuscript for typos etc and formatting issues, and done the following corrections:

- Table 1: Headings text format changed to bold and line spacing from 1,5 to 1;
N.I. changed to N. I.; homozygous to Homozygous

- Table 3: two lack of spacing typos in the last column have been corrected

Kind regards,

Robert Winqvist

Robert Winqvist, Ph.D., Professor
Laboratory of Cancer Genetics
Department of Clinical Genetics and Biocenter Oulu,
P.O. Box 5000, FIN-90014 University of Oulu, FINLAND
Tel: +358-8-3153228
Fax: +358-8-3156354
Email: robert.winqvist@oulu.fi